Author's response to reviews

Title: Prenatal diagnosis of fetal cardiac malposition caused by unilateral lung agenesis

Authors:

Ying Zhang (baogoubei@hotmail.com)
Wei-Dong Ren (renweidong_us@hotmail.com)
Li-Mei Xie (xielimei@hotmail.com)
Chang-Wei Ding (dingchangwei_r@hotmail.com)
Wei Sun (sunwei_us@hotmail.com)
Yu Wang (wangyu_us@hotmail.com)
Ya-Jun Guo (guoyajun_us@hotmail.com)
Ai-Lu Cai (caial_us@hotmail.com)

Version: 2 Date: 24 May 2012

Author's response to reviews:

Dear Editor,

We have revised our manuscript according to the requirements of the editorial staff and clarified that we had obtained written informed consents from the parents for publication of clinical details, clinical images and videos. In detail, the declaration was added in the text belongs to the subtitle “Ethics approval” in “Method” and “Consent” at the end of the manuscript. Thank you very much.

Sincerely yours,

Ying Zhang
2012-5-23